Answering Reviewers

To reviewer #1:
The authors should provide more information about this patient, such as more CT images and clinical data.
------ Unfortunately, we performed CT scan only twice. On the other hand, we added clinical and laboratory data in OUTCOME AND FOLLOW-UP section of the manuscript.

To reviewer #2:
You are stating that the patient developed a cell-mediated immune deficiency during COVID-19 treatment. Can you please add the medical evidence for the statement?
------ The following description is mentioned in some article: as COVID-19 patients have inadequate immunity and there is a possibility of reduced virus clearance after treating dexamethasone, the patient can be more immune-compromised if dexamethasone is used. We added that in discussion section of the manuscript.